

Received 7 December 2024, accepted 25 December 2024, date of publication 26 December 2024, date of current version 6 January 2025. *Digital Object Identifier* 10.1109/ACCESS.2024.3523437

RESEARCH ARTICLE

Mapping the Landscape of AI-Driven Human Resource Management: A Social Network Analysis of Research Collaboration

MEHRDAD MAGHSOUDI^{®1}, MOTAHAREH KAMRANI SHAHRI^{®2}, MEHRDAD AGHA MOHAMMAD ALI KERMANI^{®2}, AND RAHIM KHANIZAD^{®2}

¹Department of Industrial and Information Management, Faculty of Management and Accounting, Shahid Beheshti University, Tehran 19839-69411, Iran ²Department of Management, Economics and Progress Engineering, Iran University of Science and Technology, Tehran 13114-16846, Iran Corresponding author: Mehrdad Agha Mohammad Ali Kermani (M_Kermani@iust.ac.ir)

ABSTRACT As artificial intelligence (AI) transforms human resource management (HRM), understanding the research landscape becomes crucial for both academics and practitioners. While existing studies examine isolated aspects of AI in HRM, a comprehensive analysis of collaboration patterns and emerging themes remains lacking. This research employs social network analysis (SNA) to examine the co-authorship network within AI applications in HRM research, providing insights into collaboration dynamics and identifying key research directions. Through analysis of centrality measures and application of the TOPSIS method, the study identifies influential authors, institutions, and emerging research themes. Analysis of 102,296 authors and 287,799 collaborations reveals distinct communities focusing on specific aspects of AI-HRM across regions. The findings identify four primary research themes: AI for System Identification and Control, focusing on workforce planning and adaptive management; HR Analytics and Performance Management, emphasizing data-driven decision making; Machine Learning for Classification and Prediction, addressing talent acquisition and retention; and AI-Driven HR Decision-Making, exploring strategic planning and unbiased evaluation systems. The country co-authorship network analysis uncovers three main communities: Global HR Applications, HRM in the Middle East and Asia, and Global Integration of AI in HRM, reflecting shared regional challenges. Institutional collaboration patterns indicate five distinct communities, from established Asian AI research centers to emerging research hubs in developing economies. These findings provide valuable insights for researchers exploring collaboration opportunities, practitioners implementing AI solutions, and policymakers developing strategic frameworks for AI adoption in HRM. This research contributes to understanding the evolving landscape of AI-HRM research and offers practical guidelines for leveraging AI in HR practices.

INDEX TERMS Artificial intelligence (AI), human resource management (HRM), social network analysis (SNA), co-authorship network, research collaboration.

I. INTRODUCTION

The human resources (HR) function is undergoing a profound transformation, driven by emerging economic, political, social, and technological trends. Increasingly, HR is shifting from a primarily administrative role to that of a strategic business partner, essential for achieving organizational excellence

The associate editor coordinating the review of this manuscript and approving it for publication was Ramoni Adeogun^(D).

and maintaining competitive advantage [1], [2]. Strategic human resource management (SHRM) involves aligning HR practices and policies with organizational goals to enhance long-term success. It entails HR's role in business strategy by concentrating on enhancing human capital, organizational culture, leadership, and employee engagement to drive performance and innovation [3]. While traditional HR practices have focused on recruitment, selection, performance management, and retention, there is now a broader emphasis on strategic responsibilities, including fostering interpersonal interactions within organizations [4].

Artificial Intelligence (AI) has become a pivotal technology in this transformation, automating routine administrative tasks and significantly enhancing HR operations. By facilitating knowledge sharing and collaboration, AI is improving employee experiences and increasing HR effectiveness [5], [6]. However, to fully leverage AI's transformative potential, it is crucial to identify and develop the capabilities and competencies required for managers to successfully implement and oversee AI-driven HR initiatives [7].

Despite growing interest in AI within HR research, a substantial gap remains in the literature regarding comprehensive analyses of AI's impact on Human Resource Management(HRM) [8], [9]. Although AI's potential to revolutionize HRM is well-recognized, particularly in decision-making through data-driven algorithms, there is a pressing need for deeper insight into how AI can be integrated into HR subfunctions [10], [11]. Additionally, research has yet to fully explore how AI-enabled HRM functions can be aligned with other operational tasks to enhance overall employee outcomes [7], [12].

Recent studies have begun to examine AI applications in HRM, yet there remains a scarcity of research that provides a thorough overview of trends in this field [13], [14]. This paper aims to address this gap by offering a comprehensive analysis of AI's role in HRM through a novel methodologysocial network analysis (SNA). Unlike previous studies that relied on bibliometric analysis, this research employs SNA to uncover the complex relationships within AI applications in HRM. By analyzing co-authorship networks, this study reveals overlooked connections among authors, institutions, keywords, and countries, identifying emerging topics and potential knowledge hubs previously missed in the literature. Social Network Analysis is particularly valuable as it allows for an in-depth examination of the interactions and collaborations within social networks. By identifying key figures and influential entities, SNA helps to understand how network connections influence behavior, beliefs, and performance among groups and individuals [15], [16]. In the context of AI in HRM, SNA provides insights into the dynamics of author collaborations, highlighting influential contributors and their roles in advancing research. Furthermore, by mapping contributions from notable institutions and countries, this study fosters global cooperation in AI-driven HRM research.

This study provides a unique contribution to the AI-HRM research landscape by combining Social Network Analysis (SNA) and TOPSIS to map research collaboration patterns and emerging themes. Unlike previous studies, which have often focused on isolated aspects of AI in HRM, this research offers a comprehensive approach that identifies key authors, institutions, and research hubs globally. The novelty lies in the integration of network analysis with decision-making models, uncovering previously overlooked patterns in collaboration. The study not only advances theoretical knowledge

but also offers actionable insights for practitioners, policymakers, and academics involved in AI-driven HR practices.

This research seeks to answer the following key questions to enhance the understanding of AI's impact on HRM:

- 1. Who are the most influential authors in the HRM and AI co-authoring network?
- 2. Is the collaboration between countries in co-authorship driven by shared challenges, and what underpins scientific collaboration among universities and institutions in HRM and AI?
- 3. What underpins scientific collaboration among universities and institutions in HRM and AI?
- 4. Does the analysis of keyword networks in HRM and AI research reveal distinct and coherent patterns?

The structure of this paper is organized as follows. Chapter 2 reviews the literature. Chapter 3 outlines the research methodology. Chapter 4 presents the research findings. Chapter 5 discusses the results, providing explicit answers to the research questions, managerial implications, and limitations, and future research directions. Finally, Chapter 6 concludes the study.

II. LITERATURE REVIEW

This section reviews the literature on AI implementation within HRM and the use of SNA in researching collaboration. The review centers on the impact of AI on HR practices and the application of SNA to understand the nature of coauthorship networks. Synthesizing existing knowledge in this regard, this section intends to provide a good understanding of the role AI plays in modern HR practices and the nature of research collaboration in this field.

A. APPLICATION OF AI IN HRM

The integration of AI into HRM necessitates a comprehensive understanding of its capabilities and applications within the HR function [17]. AI, a field simulating human intelligence in machines, encompasses a range of technologies including machine learning, natural language processing, and predictive analytics [9]. When applied to HRM, AI automates processes, informs decision-making, and enhances employee experiences [18]. By analyzing vast datasets, AI algorithms generate insights, predictions, and recommendations to improve efficiency, reduce bias, and personalize HR interactions [19]. This integration marks a transformative shift, enabling data-driven decision-making, increased productivity, and improved employee satisfaction [17]. While AI offers significant advantages, challenges such as ethical considerations, data privacy, and the need for ongoing human expertise must be carefully addressed to ensure its responsible implementation and maximize its benefits [20]. AI is fundamentally transforming HRM practices [17]. AI-driven tools are reshaping talent acquisition by streamlining processes and enhancing candidate selection [21]. Automated systems are employed to efficiently filter and assess large applicant pools, identifying candidates aligned with organizational

needs [22]. Moreover, predictive analytics, leveraging historical data, forecasting candidate performance, and optimizing the hiring process [23]. In the realm of employee learning and development, AI-powered platforms offer personalized training experiences, adapting content to individual learning styles and knowledge levels [24].

AI also revolutionizes performance management by enabling continuous feedback, predictive analytics, and data-driven performance evaluations [22]. Additionally, AIpowered chatbots and sentiment analysis tools facilitate employee engagement by fostering instant communication and monitoring workplace morale [18]. Predictive analytics, driven by AI, offer valuable insights for talent retention, workforce planning, and succession management, enabling HR to proactively address organizational needs [25]. These AI applications transform HRM by fostering agility, strategic foresight, and a strong focus on employee well-being, thereby positioning organizations to thrive in the dynamic workforce landscape [26].

Strategic Human Resource Management (SHRM) traditionally encompasses six core functions: business administration, workforce planning, human capital development, compensation, labor relations, and risk management [27]. These domains are essential for achieving organizational objectives [19]. AI has the potential to enhance these HRM functions significantly [23]. For instance, in recruitment, AI supports HR managers by automating tasks such as filtering and preselection, thereby identifying candidates who best meet job requirements [19], [28]. Figure 1 illustrates the types and extent of AI applications in key HRM areas: Acquisition, Development, and Retention [29].



FIGURE 1. AI and strategic HRM practice choice menus [30].

AI can also significantly improve HR efficiency and quality through the use of chatbots, a common AI application in HR today [31], [32]. Table 1 outlines the potential impacts of AI on HRM in public sector organizations, both currently and in the future [33].

TABLE 1. HRM and A

HR Tasks	Potential Benefits of AI-Enabled Tools
Talent Acquisition	Increased processing capacity and more precise candidate evaluation by expanding the applicant pool
Employee Development	Immediate delivery of personalized training and onboarding
Performance Management	Comprehensive employee performance assessment, compared against peers and various metrics
Compensation	Automatic tracking of compensation across multiple data sources
Turnover and Retention	Strategic insights into incentives and potential employee departures

AI is emerging as a transformative force within HRM, with its applications extending across various HR functions [17]. For example, AI-powered tools are enhancing talent acquisition by automating candidate identification and evaluation processes [34]. Advanced algorithms analyze resumes and social media profiles to match candidates with appropriate roles, while applicant tracking systems (ATS) streamline administrative tasks, boosting efficiency [35], [36]. Job aggregators and career platforms use AI to consolidate job openings and identify potential talent pools [26], [37].

Beyond recruitment, AI is reshaping employee development by supplementing traditional training methods with AI-driven learning platforms [38]. These platforms offer personalized, data-driven learning experiences by analyzing vast datasets to identify training needs, measure learning outcomes, and optimize training programs [18], [26].

Analyzing co-authorship networks through Social Network Analysis (SNA) is vital for understanding academic collaborations, trends, and influential experts. SNA quantifies collaborative intensity, identifies central figures, and visualizes connectivity within research networks. Introduced in the mid-1900s and popularized by "small world" experiments, SNA is crucial for comprehending complex social phenomena, assessing network characteristics, and mapping collaboration landscapes in various industries, including research [39].

B. RELATED WORKS

The Related Works section focuses specifically on reviewing papers that have conducted comprehensive and systematic analyses of AI applications in HRM research. Rather than examining all individual papers exploring various aspects of AI in HRM, this section analyzes those studies that, similar to the present research, have undertaken collective reviews of AI-HRM literature. This approach provides a meta-level understanding of how researchers have systematically analyzed and synthesized knowledge in this field. The selected papers represent significant contributions in systematically reviewing and analyzing multiple studies, rather than individual implementations or specific applications of AI in HRM.

Research on AI applications in HRM has rapidly increased in recent years, enhancing HR functions and organizational outcomes. Reference [21] identified a significant lack of indepth, organized research on HRM (AI) in current studies, particularly in the development of a multilevel framework that could serve as a foundation for future research by scholars. To address this disparity, they suggest a comprehensive framework that provides a foundation for upcoming researchers to establish connections between various variables, beginning at the contextual level and leading to HRM and organizational outcomes that ultimately improve the operational and financial performance of organizations.

AI and AI-based applications are increasingly used in HRM in both local and global organizations, leading to new research on topics like the social presence of AI, its effects on outcomes, and evaluating practices. Further research is needed for a comprehensive understanding and future directions. Reference [30] present a systematic review of the theme of this special issue and offer a nuanced understating of what is known, yet to be known, and future research directions to frame a future research agenda for international HRM.

Research on AI in HRM has gained attention, but few studies address the trends and competencies needed for its adoption. Reference [24] have a systematic review and bibliometric analysis identified managerial capabilities required for AI adoption in HRM, using the Dynamic Capabilities View. Recruitment and selection are common areas for AI use. Managerial cognitive, human, and social capital are important for AI adoption. Multiple methods were used to explore AI adoption in HRM.

Next research examines how AI has revolutionized HR functions, from talent acquisition to performance management. The study utilizes a bibliometric approach to analyze research trends between 2013 and 2022, focusing on identifying key research clusters and geographical patterns.

By employing VOS viewer, the authors mapped out the intellectual structure of the field, revealing India as a leading contributor and highlighting areas such as machine learning for resource management, AI for recruitment, and AI-driven training as emerging research frontiers [40].

HRM is changing its focus from traditional methods like recruitment, training, and development to more advanced techniques like automation, AI integration, and augmented intelligence. Reference [41] explore the application of AI in different activities associated with HRM. The research identifies the particular AI technologies utilized in HRM, investigates the advantages and difficulties of incorporating AI in HR, and the strategies applied to address these hurdles. The incorporation of AI in HRM is a new trend and only a small number of businesses have successfully adopted it in every HR process.

By systematically reviewing existing literature, the research explores how AI and automation are reshaping Human Resource Development (HRD) practices. The study identified key HRD processes impacted by these technologies and analyzed their influence on overall HRD outcomes. Findings offer insights into the challenges and opportunities presented by AI and automation, contributing to the ongoing discourse on the future of HRD [42].

This research employs Gephi, a specialized social network analysis tool, for analyzing co-authorship networks. While various software packages exist for network analysis, Gephi offers distinctive analytical capabilities crucial for this study. The software provides robust computation of multiple centrality metrics (degree, betweenness, closeness, and eigenvector centrality) to identify key actors in complex networks. Its advanced community detection algorithms, combined with unique filtering features, enable detailed analysis of specific communities and their network metrics - a capability particularly valuable for understanding research group dynamics.

These specialized features facilitate comprehensive examination of scholarly collaboration in the AI-HRM domain.

Year	Research Topic	Research Method	Main Research Result
2023	Al-augmented HRM: Antecedents, assimilation and multilevel consequences	Literature review, Framework development	Lack of in-depth research on HRM (AI), Proposed a comprehensive framework
2023	AI-assisted HRM: Towards an extended strategic framework	Systematic review	Understanding of current knowledge, Future research directions
2023	Al in HRM: case study analysis. Preliminary research	Systematic review, Bibliometric analysis	Managerial capabilities for AI adoption, Common areas for AI use
2024	AI Applied to HRM: A Bibliometric Analysis	Bibliometric analysis	The study revealed a significant increase in research on AI applications in HRM
2024	Application of AI in HRM: a conceptual framework	Exploratory research	Al technologies in HRM, Advantages and difficulties of Al integration
2024	AI and Automation in HRD: A Systematic Review	Author citation and co-citation analysis, Communication analysis, Exponential random graph model	Key findings include how AI and automation influence specific HRD processes and their overall effects on HRD functions.

TABLE 2. Summary of the past research.

Through co-authorship pattern analysis, the research identifies key contributors within specific research communities, reveals knowledge flow patterns, and maps emerging research clusters. The ability to filter and analyze specific communities while computing their distinct network metrics provides deeper insights into collaboration patterns. By examining author relationships, institutional connections, and collaboration dynamics, this analysis offers a thorough understanding of the field's structure and evolution. Table 2 provides a summary of the topic, method, and results of past research.

C. INNOVATIONS IN METHODOLOGY COMPARED TO EXISTING RESEARCH

This study introduces several key innovations in the analysis of AI in HRM, which set it apart from previous works in the field:

1) INTEGRATION OF SOCIAL NETWORK ANALYSIS (SNA) WITH TOPSIS FOR RESEARCH COLLABORATION MAPPING

Previous studies have largely relied on bibliometric methods to analyze the AI-HRM research landscape. This study innovatively combines Social Network Analysis (SNA) with the TOPSIS decision-making method, providing a dual approach that not only identifies the structural patterns in research collaborations but also ranks institutions and authors based on their research influence. This combination allows for a deeper, more nuanced understanding of research dynamics compared to traditional bibliometric methods.

2) GLOBAL MAPPING OF AI-HRM RESEARCH NETWORKS

While many studies examine AI in HRM from a regional or thematic perspective, this research offers a global mapping of research collaborations. By analyzing the co-authorship networks across different regions and countries, the study reveals global research hubs and trends that have not been fully explored in the existing literature. This global perspective is a significant methodological advancement, offering insights into international cooperation in AI-driven HRM research.

3) IDENTIFICATION OF KEY RESEARCH THEMES USING SNA Another methodological innovation is the use of Social Network Analysis to identify emerging research themes. While traditional methods focus on literature reviews and thematic analysis, this study applies SNA to uncover hidden patterns in the research community, identifying themes based on collaboration networks rather than just keywords or citations. This approach reveals four primary themes of AI in HRM, offering a comprehensive view of the field's evolution and potential future directions.

4) ENHANCED QUANTITATIVE ANALYSIS WITH MULTIPLE CENTRALITY MEASURES

Unlike previous studies that typically focus on one or two centrality metrics, this research employs multiple centrality measures (degree, betweenness, closeness, and eigenvector) to assess the influence of authors and institutions. This approach provides a more comprehensive and granular understanding of the research landscape, offering insights into not only who the most influential authors are but also how they contribute to shaping the field.

5) COMPREHENSIVE DATASET AND CROSS-METHOD INTEGRATION

By integrating a large dataset of over 100,000 authors and 287,000 collaborations, and combining quantitative methods like SNA with decision-making tools such as TOPSIS, this study offers a unique methodological framework. It enables a more accurate and holistic analysis of AI-HRM research trends than previous studies that have focused on smaller datasets or singular methods.

In summary, the methodological innovations introduced in this study—such as the integration of SNA with TOPSIS, the global mapping of research networks, the identification of emerging research themes using SNA, and the use of multiple centrality measures—mark significant advances in the field. These innovations provide a deeper and more nuanced understanding of the AI-HRM research landscape compared to previous works that primarily relied on bibliometric approaches or were limited in scope. The combination of these methods enables a more comprehensive, multidimensional analysis of research collaborations, offering valuable insights into the evolution and future directions of AI applications in HRM.

III. METHODOLOGY

This scholarly article investigates the phenomenon of co-authorship within the intersection of AI and HRM. The research commenced by systematically gathering articles related to "AI applications in HRM" using a carefully curated set of keywords specific to AI and HRM [17]. The selection process involved the use of two distinct groups of keywords: one set focusing on AI-related terms and another on HRM-related terms, ensuring a comprehensive search across both domains [43]. These keywords are detailed in Figure 4 of the manuscript.

A strategic decision was made to utilize the Web of Science (WOS) as the primary database for this study. WOS was chosen due to its extensive coverage of high-quality, peer-reviewed research across disciplines such as business, computer science, and management [44]. The rigorous inclusion criteria, extensive citation indexing, and global reach of WOS provide a robust foundation for analyzing the complex relationships between AI and HRM [37]. By employing WOS, the research accesses a rich dataset that facilitates detailed analysis and the identification of significant trends and patterns within the field [17]. Many researchers have utilized this source for examining and analyzing co-authorship networks, as evidenced by studies such as [45], [46], [47], [48], and [49].

Furthermore, the study predominantly relies on the Web of Science Core Collection, a prestigious resource recognized for cataloging significant scientific inquiries and widely



FIGURE 2. Methodological framework.

respected in academic circles for its reliability and depth [50]. The research methodology, including the procedural steps for data collection and analysis, is thoroughly outlined in Figure 2 of the manuscript, ensuring transparency and replicability.

In the realm of data preprocessing, a dataset is formulated to facilitate the examination of the co-authorship network involving authors, universities, countries, and keywords. The process of dataset creation is illustrated in Figure 3 showcasing the establishment of links between authors when they collaborate on an article, with the weight of these links escalating as authors collaborate on multiple articles.

A. SOCIAL NETWORK ANALYSIS (SNA)

Analyzing co-authorship networks is now a crucial method for examining how scholars collaborate with each other. Through the utilization of Social Network Analysis (SNA) methods, scholars can acquire more profound understandings of the formation and evolution of academic collaborations. This method is especially important in the current research environment, where it is vital to grasp new trends, recognize influential experts, and forecast upcoming research directions [52]. These insights are critical for grasping how knowledge is disseminated, how research communities form, and for pinpointing potential collaborators [53].

SNA further allows for the quantification of collaborative intensity, identification of central figures, and evaluation of overall connectivity within a research network. Visualization of these networks provides a comprehensive view of the



FIGURE 3. Network formation rules [51].

research landscape, helping scholars to identify new trends and potential opportunities for collaboration [54], [55].

Social Network Analysis, first created in the mid-1900s, grew in popularity thanks to Stanley Milgram's "small world" experiments. Since gaining popularity through the "Six Degrees of Kevin Bacon" game, social network analysis (SNA) has developed into a flexible approach applied in various industries including healthcare, marketing, national security, political science, and particularly, research collaboration studies [56]. Essentially, SNA focuses on examining the connections between elements, depicted as nodes and edges in a network graph. This method enables researchers to discover important figures, reveal obscured connections, and analyze the distribution of resources or information in a system. SNA is a valuable method for comprehending intricate social occurrences, from illness spread to corporate conduct [57], [58].

Besides recognizing network elements, social network analysis provides numerical measurements to assess network characteristics. Indicators like density, clustering coefficient, and centrality offer important information about the unity, interconnection, and importance of nodes in a network [59]. These metrics enable researchers to compare networks, detect structural differences, and draw meaningful conclusions about the systems under investigation. In research collaboration, SNA helps map the co-authorship landscape, revealing key scholars, emerging research topics, and potential collaborators. Moreover, SNA is instrumental in uncovering hidden knowledge networks and fostering interdisciplinary collaboration [60].

Additional metrics such as community detection, density, and clustering coefficient provide further clarity into the overall characteristics of a network [61]. These metrics help in identifying cohesive subgroups, assessing overall connectedness, and understanding the distribution of collaboration patterns within the field of research. By applying these SNA techniques, researchers can better understand collaboration dynamics, pinpoint emerging trends, and develop strategies for knowledge dissemination and impact [62].

Social Network Analysis (SNA) plays a pivotal role in examining network structures through a wide array of metrics designed to quantify relationships between entities, evaluate the centrality of key actors, and identify cohesive subgroups within the network [63]. The following metrics are integral for researchers seeking to understand network dynamics and inform subsequent analyses [64], [65]:

- **Degree:** The number of connections a node has is referred to as its degree, representing the size of node iii's neighborhood. A high degree of centrality is typically observed in active individuals who maintain numerous connections with other nodes within the network. This reflects a node's influence and centrality within the system [66], [67].
- **Closeness:** Closeness centrality measures a node's proximity to other nodes in the network by calculating the average distance to every other node [68]. A node with high closeness centrality is positioned near other nodes with shorter paths, indicating its capacity to quickly access or disseminate information across the network. In a literary context, authors with higher closeness centrality tend to be more closely connected to other authors, signaling higher visibility and influence within the scholarly community [69].
- Betweenness Centrality: Betweenness centrality quantifies the frequency with which a node lies on the shortest path between any two nodes in the network [70], [71].

Nodes with high betweenness centrality often act as critical intermediaries or bridges, playing a pivotal role in controlling the flow of information and content throughout the network.

- Eigenvector Centrality: Research suggests that the top eigenvector of an adjacency matrix can be an effective indicator of network centrality [51]. Unlike degree centrality, eigenvector centrality assigns varying weights to nodes based on their connections, emphasizing the importance of influential neighbors. Eigenvector centrality provides a weighted combination of direct and indirect links, capturing the influence of both short and long-range connections within the network [72].
- **Density:** The density of a network with nnn nodes is determined by the ratio of the actual number of edges (mmm) connecting these nodes to the maximum potential number of edges in the network. This measure reflects the proportion of possible connections that are currently realized, offering insights into the network's overall connectivity [73], [74].
- **Community Detection:** Identifying communities is a key aspect of Social Network Analysis (SNA), facilitating an understanding of the structure and behavior of networks. A community is a structural element that showcases the relationship between nodes, with the detection process beginning by forecasting connections between nodes to identify more densely connected clusters. This process helps researchers better understand the underlying structure and dynamics of the network, aiding in the identification of cohesive groups that exhibit strong internal ties while maintaining weaker connections with other groups [75], [76].
- Size: number of the network's nodes [77].

One of the most common ways to analyze a network is to look at the centrality of various nodes to identify key players, information hubs, and gatekeepers across the network. There are three types of centralities, each corresponding to a different aspect of connectivity and centrality. Degree, Betweenness, and Closeness Centrality are measures of a node's importance. Table 3 delineates several centrality indicators, including Degree, Closeness, and Betweenness centralities [78].

The objective of the network formation phase is to establish a co-authorship network spanning authors, universities, countries, and keywords. Utilizing the datasets generated in the preceding phase, Gephi software is employed to generate and visualize the networks. Widely recognized as a prominent tool for social network analysis and visualization, the Gephi 0.9.7 software version prepares us for the subsequent scrutiny of the networks.

The fourth phase involves the examination of the developed networks. This examination encompasses the application of centrality metrics such as Eigenvector, Degree, Closeness, and Betweenness, alongside the employment of community detection techniques.

Centrality measure	Description	Formula
Degree	The number of links involving that node	$\sum_{i=1}^{n} a(n_i, N_j)$
Closeness	The standard deviation of the length of the shortest path from one node to all other nodes	$\frac{N-1}{\sum_{i=1}^{N} D(j,i)}$
Betweenness	The number of times a node is found along the shortest possible path to other nodes.	$\frac{1}{N^2} \sum_{\forall s,t \in V} \frac{N_{S,t}^I}{G_{S,t}}$
Eigenvector	The sum of the centrality values of the nodes To which it is connected	$\frac{1}{\Lambda} \sum_{I=1}^{N} A_{I,j}, X_{J}$

TABLE 3. SNA measures.

B. TOPSIS

A method utilized to identify the optimal choices by considering various factors is referred to as multiple-criteria decisionmaking (MCDM). This strategy is also known as multipleattribute decision-making (MADM) and multiple-objective decision-making (MODM), as introduced by Hwang and Yoon in 1981 [79]. TOPSIS was selected for this study due to its ability to efficiently handle multi-dimensional data and offer a comprehensive ranking framework. Unlike traditional bibliometric methods, which rely on basic metrics like citation counts or co-authorship, TOPSIS provides a multi-criteria approach for ranking the influence of authors. By integrating multiple centrality measures (such as degree, betweenness, closeness, and eigenvector centrality), it enables a more nuanced evaluation of authors based on both their quantity of collaborations and their strategic position within the research network. The procedural components of the TOPSIS technique consist of the following steps:

Step 1: create a decision matrix $(Dij)_{p \times q}$ with p rows as options (nodes) and q columns as criteria.

Step 2: The dimensionless matrix $L=(lij)_{p\times q}$ is created by normalizing the decision matrix:

$$l_{ij} = \frac{D_{ij}}{\sqrt{\sum_{i=1}^p D_{ij}^2}}$$

Step 3: Each criterion's relative importance is shown by a weighted matrix T.

Calculate T as follows:

$$T = (t_{ij})_{p \times q} = (w_j l_{ij})_{p \times q} \quad \forall i, j$$

Step 4: The worst solution t_j^- and the best solution t_j^+ are defined as follows:

$$t_{j}^{+} = \{ (Maxt_{ij} | i = 1, 2, \dots, p.\forall_{j} \in J_{+}), \\ (Mint_{ij} | i = 1, 2, \dots, p.\forall_{j} \in J_{-}) \} \\ t_{j}^{-} = \{ (Maxt_{ij} | i = 1, 2, \dots, p.\forall_{j} \in J_{-}), \\ (Mint_{ij} | i = 1, 2, \dots, p.\forall_{j} \in J_{+}) \}$$

 J^+ represents the set of benefit criteria and J^- represents the set of cost criteria. A set of criteria, J^+ , is aimed to be greater, while a set of criteria, J^- , is aimed to be less.

Step 5: Distance all alternatives (options) and the best solution t_i^+ is calculated as follows:

$$S_i^+ = \sqrt{\sum_{i=1}^q (t_{ij} - t_j^+)^2}, i = 1, 2, \dots, p$$

Also, the distance between the whole alternatives and the worst-defined solution t_i^- is calculated as follows:

$$S_i^- = \sqrt{\sum_{i=1}^q (t_{ij} - t_j^-)^2}, i = 1, 2, \dots, p$$

Step 6: Defining the comparative closeness to the best solution as follows:

$$C_i = \frac{S_i^-}{S_i^- + S_i^+}$$

Step 7: Alternatives are ranked according to C_i

IV. RESULT

A. DATA GATHERING

The investigation collected a significant amount of information on the role of AI in HRM studies. To specifically identify articles related to "AI in HRM," the search was initiated using a carefully selected set of keyword search terms, as illustrated in Figure 4. These terms cover topics pertinent to HRM alongside those relevant to AI. The choice of the Web of Science (WOS) database was based on its extensive coverage and advanced search capabilities, which are particularly well-suited for academic research. WOS enables the application of search filters to specific fields such as the title, abstract, and keywords. This functionality is crucial for refining search results and minimizing the inclusion of irrelevant articles, thereby improving the focus and relevance of the retrieved articles. Many researchers have extensively utilized WOS in their studies, highlighting its value in academic investigations across various disciplines [47], [48], [51], [80]. The detailed keyword strategy and the types of extracted documents are depicted in Figure 4.

The initial exploration resulted in 32,103 publications based on the selected keywords. Subsequently, the team of researchers scrutinized the distribution of these publications based on their respective publication years and illustrated the outcomes in Figure 5. The data indicates a steady rise in the quantity of publications over time, reflecting a heightened interest in the subject among scholars. The complete dataset is accessible via the link provided.¹

B. AUTHORS NETWORK

Recognizing the important participants in the co-authorship network is essential for understanding their notable role in scientific productivity and research cooperation within the network. According to [78], these individuals are instrumental in the network's general research efforts. Figure 6 provides a visual representation of the co-authorship network, where the darker, denser central area highlights the core group of

¹https://zaya.io/AIinHRM

Keywords earch Term Human Resource Management OR HR Management OR HRM Practices OR Strategic HRM OR Sustainable HRM OR Employee Performance OR Employee Turnover OR Job Satisfaction OR Global Talent Management OR HRIS OR AI in HR OR Green HRM OR Employee Relations OR HR Risk Step1: OR Employee Resilience OR Human Capital OR HR Flexibility OR Collaborative HRM OR Innovative HR Strategy AND Artificial Intelligence OR Machine Learning OR Deep Learning OR Social Network Analysis OR Text Mining OR Neural Networks OR Natural Language Processing OR Computer Vision OR Data Mining OR Linear Regression OR Logistic Regression OR Decision Tree OR Naive Bayes OR Support Vector Machine OR KNN OR K-Means OR Random Forest OR Dimensionality Reduction OR Gradient Boosting OR AdaBoost OR Reinforcement Learning OR Supervised Learning OR Unsupervised Learning OR Semi-Supervised Learning OR Ensemble Learning OR PCA OR Clustering OR GBM OR XGBoost OR LightGBM OR CatBoost OR SVM OR LSTM OR BERT OR Business Intelligence OR Expert Systems Database Types of Documents Collected Data Gatherir Article 30245 Step2: **Book Chapter** 5

Proceedings Paper

Review

FIGURE 4. Search strategy and results.



FIGURE 5. Year of publication.

researchers with the most connections. The lighter, sparser outer areas show fewer connected nodes, possibly belonging to different clusters or communities within the network. This graph represents a social network to a biological network depending on the field of AI in HRM.

Table 4 provides a detailed overview of the co-authorship network in the field of AI in HRM. This network analysis offers valuable insights into the structure and dynamics of collaboration within this domain.

Figure 6 consists of 102,296 individual entities (nodes), each representing an author in the AI in HRM field. With 287,799 edges, there are nearly three times as many connections as there are nodes. This high ratio of edges to nodes indicates significant collaboration among researchers in this domain. The average degree of 5.627 suggests that each author typically collaborates with about 5 or 6 others, pointing to a tendency for small to medium-sized research groups in this field.



769

1084

FIGURE 6. AI in HRM co-authorship network.

The network diameter of 25 indicates that the furthest two authors are separated by 25 steps or connections. This considerable diameter may reflect the existence of multiple specialized sub-domains within AI in HRM that might have limited interaction. The graph density of 0.0001 shows a very sparse network, meaning many potential connections between authors in this field do not exist. This sparsity could indicate opportunities for increased collaboration in the future.

The presence of 14,211 connected components suggests the network comprises many small, isolated clusters of authors. This structure could indicate specialization in various sub-areas of AI in HRM. The high average clustering coefficient of 0.956 further supports this interpretation, indicating that authors tend to form tightly-knit groups or communities. This suggests strong collaboration within specific research teams in the AI in HRM domain.

The dense black center of Figure 6 likely represents a highly connected core where leading and prolific authors in AI in HRM are closely linked to one another. This central group of researchers potentially plays a key role in advancing the field and bridging different research communities.

TABLE 4. Details the co-authorship network.

Metric	Value
Nodes	102,296
Edges	287,799
Average Degree	5.627
Network Diameter	25
Graph Density	0.0001
Connected Components	14,211
Avg. Clustering Coefficient	0.956

The collaborative practices of individual authors in the field of AI in HRM were assessed using centrality measures to identify those with the highest popularity, centrality, and prominence in the co-authorship network. The authors with the most connections to other authors in the network were identified using centrality degree, and Table 5 suggests amalgamating centrality metrics using TOPSIS, a technique for multi-attribute decision-making [78] and the authors with the highest centrality in AI in HRM. This can be utilized to measure the extent of trust and the level of collaboration among authors. Pedrycz Witold and Chen, C.L. Philip, had the most robust connections, suggesting that they collaborated most often on these publications.

Pedrycz Witold has been identified as the author who collaborated the most with others in the field of AI in HRM, with Chen, C.L. Philip, and Herrera Francisco following closely behind. These writers were seen as the most powerful and involved partners in the network, having the greatest number of connections to other authors in the AI in HRM field due to their high degree.

Authors with a high betweenness centrality are seen as vital in linking various author groups, as per social network analysis. This promotes interaction and cooperation among typically unconnected groups. Pedrycz Witold had the top betweenness centrality in the AI in HRM co-authorship network, functioning as an intermediary or connector. Chang, Wb possessed the greatest closeness centrality, demonstrating substantial impact within the network. Writers such as Chen, C.L. Phillip, and Wang, Wei played significant roles in the network as well. When new AI subjects or uses are introduced, they are often imitated by other authors such as Chang, Kia-Yi, and Juang, Jg.

Pedrycz Witold served as a facilitator in the collaboration network of AI in HRM. His ties to the other writers were restricted.

In Table 6, Pedrycz Witold showed the highest eigenvector centrality, indicating he is the most influential person in the network and likely collaborated with other important individuals.

Selecting the crucial and influential nodes poses a challenge due to the abundance of metrics and outcomes to consider.

Following the identification of the top 20 authors using TOPSIS, we explored the areas of expertise of these authors which are shown in Table 6, and found that they work in data analysis, with a particular interest in HR, a trend that has been growing in recent years. This enabled us to determine the top authors' majors and areas of study.

Out of the nine countries listed in the table of the top 20 most central researchers and their countries, China is represented by five authors. England boasts four authors, Spain has three, while Canada and the United States each have two, with countries such as India, Singapore, the Netherlands, and Australia having just one representative each Table 7 offers a thorough compilation of each country and the number of writers represented. This table displays the top countries of the most successful authors within the AI and HR field.

Studying the scientific connections established between various countries is a useful and intriguing analysis in the co-authorship network. The visual representation of these publications is shown in Figure 7 illustrating the country co-authorship network.

Research on the intersection of AI and HRM has been conducted by scholars worldwide. A network analysis reveals collaborations among 127 countries through 1524 links. China and the United States are prominent collaborators, followed by Canada and China, then the United States. Leading nations in collaborations include China, the United States, Australia, England, and India. Community detection in the network shows close collaborations among countries in the field of AI in HRM research.

The green population is the largest. The green community consisted of 69 countries, with England, France, and Spain being the top three in AI in HRM research. However, the most efficient country (USA, China, Canada) is part of the Pink Community, which comprises 32 countries.

Table 8 analyzes the country's co-authorship network and includes three Communities.

In the expansive world of global HR, three distinct communities have emerged, bringing diverse viewpoints and valuable insights on effectively incorporating AI into HR operations. As companies adjust to the constantly changing HR environment, these communities serve as centers for creativity, collaborating to shape the future of managing employees by incorporating advanced AI technologies.

TABLE 5. Most central authors.

Rank	Degree	Closeness	Betweenness	Eigenvector
1	Pedrycz Witold	Chang, Wb	Pedrycz Witold	Pedrycz Witold
2	Chen, C.L.Philip	Juang, Jg	Chen, C.L.Philip	Chen, C.L.Philip
3	Herrera Francisco	Chang, Kia-Yi	Wang, Wei	Wang, Wei
4	Wang, Wei	Tai, Wei-Shen	Herrera Francisco	He, Wei
5	Herrera-Viedma, Enrique	Chang, HH	Ding, Weiping	Liu, Yang
6	He, Wei	Yalcin, Haydar	Yang, Jie	Yang, Chenguang
7	Wang, Jing	Aldhaban, Fahad	Liu, Yang	Su, Chun-Yi
8	Liu, Yang	Melovic, Boban	He, Wei	Herrera-Viedma, Enrique
9	Shi, Peng	Harmon, Robert	Martinez, Luis	Shi, Peng
10	Wang, Zidong	Cirovic, Dragana	Yao, Xin	Wang, Shitong
11	Nie, Feiping	Basoglu, Nuri	Wang, Zidong	Shi, Peng
12	Li, Tianrui	Veljkovic, Slavica Mitrovic	Wang, Yu	Wang, Zidong
13	Wang, Yu	Vulic, Tamara Backovic	Shi, Peng	Lin, Jerry Chun-Wei
14	Wang, Shitong	Alassaf, Deemah	Yang, Bo	Herrera, Francisco
15	Wang, Jun	Ahrens, Diane	Li, Tianrui	Wang, Shitong
16	Li, Hui	Yucelbas, Cuneyt	Hu, Qinghua	Wang, Jun
17	Wang, Hao	Shifter, Dara	Chen, Hao	Wu, Min
18	Jiao, Licheng	Wohlrab, Lars	Kaynak, Okyay	Wang, Yu
19	Li, Yan	Smirnov, Evgueni N.	Yang, Chenguang	Yun, Unil
20	Hong, Tzung-Pei	Sprinkhuizen-Kuyper, Ida G	. Wang, Jun	Chen, Long

These three communities set themselves apart by focusing on different aspects:

- *The Global HR Applications:* This community emphasizes the influence of European countries such as England, Spain, France, Germany, and Italy in the field of Global HR. The joint efforts of participants from these countries show a common commitment to incorporating AI into HR. The participation of these European nations showcases the broad global representation in the community, aiding in a comprehensive evaluation of the impact of AI on HR practices globally. As key members of the European Union, these countries influence EU decisions concerning AI and HRM, impacting the global stage with their actions.
- HRM in The Middle East and Asia: This section focuses on the leading role of Malaysia, Pakistan, Egypt,

the United Arab Emirates, and Jordan in implementing innovative HR practices within emerging markets. Writers from these economies are playing a significant role in influencing HR strategies, especially with the increasing presence of AI and the demand for creative HR administration. These nations need further study because of their expanding economies, with HR practices being shaped by both government rules and societal beliefs. Of great geopolitical importance, they are essential for trade and economic interactions in nearby areas. By investing in infrastructure and prioritizing skill development, these countries are leveraging their young workforce and AI to improve global business practices by evolving HR strategies.

• *Global Integration of AI in HRM:* In countries such as China, the USA, Australia, India, Canada, Iran,

TABLE 6. 20 effective authors.

Rank	Author	Institute	Country	Field of study
1	Pedrycz, Witold	University Of Alberta	Canada	Faculty Of Engineering - Electrical & Computer Engineering Dept.
2	Chen, C. L. Philip	South China University of Technology	China	Computational, Intelligence Intelligent, Control Cybernetics, Intelligent Transportation Systems, Data Science
3	Wang, Wei	University of California, Los Angeles	USA	Data Mining, Machine Learning, Big Data Analytics, Bioinformatics, and Computational Biology Computational Medicine
4	Herrera, Francisco	Granada University	Spain	AI, Computational Intelligence, Data Mining, Evolutionary Algorithms, Big Data Analytics
5	He, Wei	Purdue University Northwest	Indian	Strategic Inter-Firm Relationship Management, Global Innovation, and Knowledge Management
6	Liu, Yang	Nanyang Technological University	Singapore	Cybersecurity, Software Engineering and Al
7	Wang, Zidong	Brunel University London	England	Intelligent Data Analysis, Control Engineering, Signal Processing, Bioinformatics
8	Yao, Xin	University of Birmingham	England	Evolutionary Computation, Neural Network Ensembles, Multiple Classifiers, Meta-Heuristic Algorithms, Data Mining, Global Optimization
9	Shi, Peng	The University of Adelaide	Australia	Systems And Control, Electrical And Electronic, Engineering, Cybernetics, Human-Computer Interaction
10	Yang, Jie	Delft University of Technology	Netherlands	Human-Centered AI, Crowd Computing, Human-Centered Computing, Human Language Technologies, Recommender Systems
11	Su, Chun-Yi	Concordia University	Canada	On holonomic Mechanical Systems, Mechatronics, Fuzzy Control Techniques
12	Martinez, Luis	University Of Jaén	Spain	Fuzzy Decision Making, Computing with Words, Recommender Systems
13	Yang, Chenguang	University Of the West Of England Bristol	England	Neural Networks and Learning Systems, Robotics
14	Wang, Yu	University Of Hong Kong	China	Analytic Of Geo-Data, Geotechnical Risk and Reliability, Probabilistic Site Characterization
15	Ding, Weiping	Nantong University	China	Feature Extraction, Pattern Classification, Rough Set Theory, Image Segmentation, Decision Making
16	Herrera-Viedma, Enrique	University Of Granada	Spain	Fuzzy Sets, Fuzzy Decision Making, Computing with Words
17	Li, Tianrui	Southwest Jiao Tong University	China	Data Mining, Big Data, Cloud Computing
18	Wang, Jun	University College London	England	Machine Learning, Multi-Agent Learning, Information Retrieval
19	Wang, Shitong	Massachusetts Institute of Technology	USA	Nuclear Science and Engineering
20	Wang, Jing	University of Guangdong,	China	Data Mining, Big Data, and Educational Theory Technology

TABLE 7. Country of 20 effective authors.

Number of Effective Authors	5	4	3	2	1
Country	China	England	Spain	Canada & USA	Indiana, Singapore, the Netherlands, and Australia



FIGURE 7. Co-authorship network of countries.

Poland, Japan, Singapore, Vietnam, South Korea, and Taiwan, it is essential to have international cooperation for AI-powered workforce management. Writers from technologically advanced countries are collaborating to improve AI-based workforce management. This collaboration showcases the worldwide significance of AI in influencing contemporary workplace tactics and HR protocols. These nations give importance to diversity and inclusion in the work environment to foster inclusive atmospheres. They have crucial functions in the worldwide economy, trade, business, politics, and diplomacy. The incorporation of AI in businesses, particularly in HR management, has transformed the operational processes, enabling the creation of a diverse and inclusive workplace.

C. UNIVERSITIES NETWORK

The analysis of organizational collaboration provides valuable insights into the most efficient and influential institutions [81]. The involvement of organizations in the fields of HRM and AI has laid the foundation for an institutional coauthorship network. This network, consisting of 4,036 nodes and 24,655 links, highlights collaborative efforts in publications related to HRM and AI. The size of each node represents the total number of connections among institutions, with each node signifying a distinct institution. Research partnerships are more likely to occur when there is a stronger connection between institutions. Numerous organizations have collectively contributed to papers in the fields of HRM and AI.

This institutional co-authorship network has been established through a collective effort of numerous organizations, each contributing to the advancement of knowledge in the fields of HRM and AI. The network's density and connectivity suggest that these organizations have formed meaningful relationships, collaborating on research projects and sharing knowledge and expertise. This increased collaboration has led to a significant body of research that has shed light on the impact of AI on HRM practices and the implications for organizations.

The various colors in Figure 8 represent different communities of institutions. In the graph, inter-institutional co-authorship is indicated by "edges," while each institution is represented by a "node." The thickness of the edges illustrates the level of scientific collaboration between two nodes, with thicker edges indicating stronger cooperation. A detailed overview of the five communities is presented in Table 9.

Different approaches can be used to categorize universities and scientific institutions within the AI sector, taking into account factors such as cultural and social contexts, the developmental stage of countries, and the presence of major research hubs like those in the U.S. Based on these criteria, institutions were classified and segmented into communities, with each community named according to its classification method.

1) AI RESEARCH CENTERS ARE DOMINANT IN ASIA

AI Research Centers that are Dominant in Asia: Asia has emerged as a notable contributor to AI research, with the Chinese Academy of Sciences taking the lead in establishing partnerships with different research institutions in China, Hong Kong, and Singapore. The goal of this joint initiative is to further research on AI in the area [82]. Moreover, Chinese enterprises such as Lenovo, Formosa Plastics, Haier, and Huawei Technology have effectively implemented Western HRM, granting them a competitive advantage in the market [83]. The incorporation of these methods equipped them with the essential skills and knowledge for contemporary HRM practices [84]. These companies encountered difficulties in choosing and advancing staff through personal relationships with leaders, which led to the necessity of implementing competency-based selection and performance-driven incentive systems from Western HRM [85]. As Figure 9 illustrates, the adoption of these

Community color	Suggested Name	Number of edges and nodes	Top 5 country	Density
Green	The Global HR Applications	Nodes 69 Edges 475	England, Spain, France, Germany, Italy	0.202
Blue	HRM in the Middle East and Asia	Nodes 26 Edges 68	Malaysia, Pakistan, Egypt, UAE, Jordan	0.209
Pink	Global Integration of AI in HRM	Nodes 32 Edges 160	China, USA, Australia, India, Canada	0.323

TABLE 8. Communities of country co-authorship network.

Western HRM practices has significantly impacted the performance of these Chinese companies. To remain competitive and ensure long-term growth, Chinese companies need to keep incorporating leading global AI research and HRM practices as they navigate the changing business environment [86].



FIGURE 8. AI in HRM institutional co-authorship network.

2) AI IN LEADING UNIVERSITIES IN THE UNITED STATES

American universities are part of the vibrant orange community, which consists of esteemed institutions and systems in the United States. This network partnership involves a mutual emphasis on the importance of AI in higher education and research in the country [87]. The University of Florida, the University of Texas, and the University of California are prominent participants in this energetic community. These groups collaborate closely in the realms of AI and HRM [17].

The United States is experiencing distinct obstacles as a result of its economic success, which is causing a lack of workers. Creative hiring techniques are being implemented with a focus on varied backgrounds. HR professionals are

TABLE 9. Details of five communities.

Community Name	Community Color	Nodes and Edges Density
Al Research Centers that are Dominant in Asia	Purple	Nodes 664 Edges 4093 0.015
AI in leading universities in the United States	Orange	Nodes 619 Edges 2579 0.013
Al in Higher Education on a Global Scale	Green	Nodes 415 Edges 1264 0.015
Leaders in Al research in Europe	Pink	Nodes 361 Edges 981
Newly established centers dedicated to AI research	Gray	Nodes 192 Edges 643



FIGURE 9. AI research centers dominant in Asia.

acknowledging the necessity of effectively managing virtual teams [9]. HR professionals need to acquire technological skills to adjust to the evolving dynamics of the workplace. HR is transforming into a valuable partner, aligning its efforts



FIGURE 10. AI in leading universities in the United States.



FIGURE 11. Leaders in AI research in Europe.

with the objectives of the organization. Companies can gain a competitive advantage by implementing high-performance work systems, offering effective incentives, and maintaining a positive employer brand image to attract and retain talented employees. This strategic method plays a key role in the organization's overall achievement [88], [89].

American colleges are currently dealing with the task of integrating advancements in AI with HRM growth, with the orange group emerging as a hub of creativity driving progress in academia, research, and staff supervision [90]. Figure 10 demonstrates the rapid growth of AI adoption within the orange group universities. The competitive positioning of these institutions and businesses is made possible through the integration of AI technologies and strategic HR practices.

3) LEADERS IN AI RESEARCH IN EUROPE

Within the Pink community, a cluster of prominent European universities has emerged as leaders in cutting-edge AI research. The London University, Edinburgh University, and KU Leuven are among the premier institutions that feature prominently in Figure 11. The unique characteristics of each institution have converged to create a vibrant research ecosystem, where AI research is being pushed forward by talented researchers and scholars. HRM practices in Europe differ from those in the United States by being more informal. [17] states that European traditions consist of more relaxed grading systems and organizational structures. Workers in Europe favor a hierarchical management approach, whereas the U.S. offices typically rely on a system of organizing and reporting to oversee task completion.

A key difference between HRM in Europe and the United States lies in the management approach. European offices typically employ a hierarchical management structure, where decisions are made through a clear chain of command. In contrast, US offices often rely on a more decentralized system, where employees are given greater autonomy to complete tasks and projects, with regular reporting and feedback from their supervisors.

These differences in HRM practices have significant implications for AI research in Europe. The more relaxed and collaborative culture has fostered a sense of community and cooperation, allowing researchers to share knowledge and expertise, and to collaborate on projects that might not have been possible in a more formal environment. Additionally, the hierarchical management approach in European offices has enabled researchers to focus on specific areas of AI research, and to allocate resources effectively, rather than being bogged down in bureaucracy.

4) AI IN HIGHER EDUCATION ON A GLOBAL SCALE

The lively green community which is shown in Figure 12 made up mostly of Iran, Egypt, and Malaysia, is actively promoting collaboration and exchanging ideas on integrating AI into academic environments. Universities from different countries unite in this diverse group to showcase a worldwide outlook on the significance of AI in higher education [17].

A recent study has emphasized the importance of HRM practices in achieving organizational success. Utilizing HRM functions effectively can serve as a strategic advantage for companies to attain success [91]. Effectively utilizing HRM functions can serve as a strategic advantage for companies seeking to attain success. The research highlights the interconnectedness of culture and HRM, with a specific focus on Islamic HRM methods in nations like Iran, Egypt, and Malaysia. Islamic principles play a crucial role in guiding conduct, but the lack of well-defined principles can create challenges when attempting to apply Islamic HRM techniques in Muslim nations. The study identifies eight essential



FIGURE 12. AI in higher education on a global scale.

elements of Islamic HRM and proposes that incorporating these principles can be beneficial for both companies and their staff members. Furthermore, exploring international perspectives on AI in academia involves examining cultural variations in HRM tactics, acknowledging the importance of context-dependent HRM practices [25].



FIGURE 13. Newly established centers dedicated to AI research.

5) NEWLY ESTABLISHED CENTERS DEDICATED TO AI RESEARCH

Figure 13 illustrates a cluster of universities and research institutions, forming a diverse group that combines developing and progressing economies. This partnership embodies a unified effort to promote progress in AI research within these fields, highlighting the potential impact of AI on the broader international AI environment. The collaboration within this group is poised to have a profound effect on the growth and advancement of AI globally, as seen in the emergence of international efforts such as the Sino-European Laboratory of Informatics, Automation and Applied Mathematics (LIAMA) [92]. Originally established to encourage Chinese colleagues to publish in international journals, LIAMA has evolved into a collaborative partnership on the global stage, marked by equal contributions from stakeholders. This is a testament to the power of international cooperation in AI research, as well as the potential for knowledge sharing and innovation. Moreover, these centers are not only advancing AI but also addressing pressing global challenges, aligning their work with the United Nations' Sustainable Development Goals (SDGs) [93]. The AI for Social Good movement, for instance, focuses on leveraging AI to drive positive change and address pressing societal issues, such as poverty, inequality, and environmental degradation. This strategic alignment enables researchers to tackle complex problems, while also promoting sustainable development and human welfare [94].

D. KEYWORDS NETWORK

The presence of specific words within a set of articles on a given topic can reveal the primary subjects addressed in those articles. The key concepts of an article are often reflected in its keywords. Words that cluster together in documents typically relate to closely connected themes. Each cluster encompasses a different set of subject-specific keywords. By analyzing these keywords, one can discern the relationships and proximity between various topics [95].

In the context of AI & HRM research, keywords can be organized into four distinct communities, as shown in Figure 14. According to [96], keyword search analysis is a valuable method for predicting emerging scientific trends.



FIGURE 14. Keywords network.

The AI & HRM articles' keywords can be categorized into four Communities according to the results of community detection. Keyword search analysis is identified by [96] as an effective strategy for forecasting upcoming scientific trends.

A network of keywords is studied, with 176894 links and 39363 nodes. The number of nodes representing each keyword shows how often it appears in the record. Keywords are crucial terms that aid readers in finding and understanding the concepts and information presented in research papers.

A closer examination of the data within communities may lead to a clearer understanding of how words are allocated to different communities and the relationships between these communities of words. Identification of four keyword communities was achieved through community detection in the indexed corpus keywords. Table 10 provides in-depth data on the number of edges and nodes.

neural networks Neural network Adaptive control

Neural networks

FIGURE 15. Green keywords community.





the potential of these techniques, the community is seeking to uncover innovative solutions that can revolutionize HR processes. One of the primary areas of interest within this community is the development of neural networks, which can be used to model complex HR datasets and make predictions about employee behavior and performance. By leveraging adaptive control systems, HR professionals can adjust and refine their strategies in real-time, ensuring that HR processes are dynamic and responsive to changing organizational needs. System identification, another key area of focus, involves analyzing and understanding HR systems and processes to identify areas for improvement. By applying AI-powered analytics and machine learning algorithms to HR data, the community is working to develop more accurate and effective solutions for HR management. The community's emphasis on stability and resilience also highlights the importance of

TABLE 10. Details of keywords community.

	Number			
Color	of Nodes	Density	Topics	Top 5 of Community
	and Edges			
	Nodes			
Rod	4099	0 002	AI and	ML, AI, Big data, Text
neu	Edges	0.002	Big Data	mining, Prediction
	12739			
	Nodes			SNA Performance
Purnle	5734	0.001	Performance	Innovation
rupic	Edges	0.001	Management	Sustainability KM
	19490			Sustaina Sinty) hiti
				Classification,
	Nodes			Ensemble learning,
Orange	5674	0.001	Machine	Extreme learning
	Edges		Learning	machines, Genetic
	21075			algorithms, Semi-
				supervised learning
	Nodes			Neural networks,
Green	4788	0.002	AI for System	Adaptive control, Fuzzy
2.001	Edges	0.002	Identification	logic, Nonlinear system,
	17749			Uncertainty

The keyword network is divided by the use and thematic function of keywords in HR. An example would be AI for system identification and reasoning control, which holds a significant position within the green cluster of AI keywords. The next steps include adaptive control and system identification, which are crucial uses of AI. Therefore, naming the clusters involves studying the impact of keywords on HR and the interaction of the biggest nodes [14], [97].

1) AI FOR SYSTEM IDENTIFICATION AND CONTROL

The eco-friendly community, depicted in Figure 15, constitutes a collective of researchers and experts who share a passion for exploring novel applications of AI in HR. The community's focus on neural networks, adaptive control, and system identification underscores the significance of these areas in implementing AI in HR management. By exploring creating HR systems that are robust and able to withstand the complexities and uncertainties of modern organizations. By developing AI-powered HR systems that are designed to be stable and resilient, HR professionals can ensure that HR processes are efficient, effective, and able to adapt to changing circumstances.

Overall, the eco-friendly community's research and development efforts aim to create a new generation of AI-powered HR management systems that are more efficient, effective, and responsive to the needs of modern organizations. By exploring the potential of neural networks, adaptive control, and system identification, the community is working to redefine the future of HR management and create a more sustainable, equitable, and prosperous work environment for all [98].

2) HR ANALYTICS AND PERFORMANCE MANAGEMENT

The purple group Figure 16 has identified key areas of focus in HR, prioritizing performance, innovation, sustainability, knowledge management, and corporate social responsibility. Within these areas, HR Analytics and Performance Management play a crucial role. By leveraging HR analytics, organizations can make data-driven decisions, optimize HR processes, and drive business outcomes. Meanwhile, performance management is essential for setting goals, measuring progress, and ensuring accountability. By aligning these areas, HR professionals can create a more effective, efficient, and strategic HR function that supports the organization's overall objectives.

The purple group's focus on small and medium-sized enterprises (SMEs) is particularly noteworthy, as it highlights the importance of utilizing AI to enhance and simplify HR processes in companies of all sizes. By leveraging AIpowered HR solutions, SMEs can streamline tasks, reduce administrative burdens, and make more informed decisions. Moreover, AI can help SMEs to stay competitive in the marketplace, attract and retain top talent, and improve employee engagement. By focusing on these key areas, the purple group is poised to make a significant impact on the HR industry, driving innovation, and shaping the future of HR management for years to come [99].

3) MACHINE LEARNING FOR CLASSIFICATION AND PREDICTION

The orange community focuses in Figure 17 is dedicated to exploring the applications of machine learning techniques in various domains, including classification, regression, and ensemble learning. These methods have far-reaching implications in HR management, enabling organizations to make informed decisions, identify trends, and predict outcomes. For instance, classification techniques can be used to identify high-potential staff, while regression methods can help predict employee turnover. Ensemble learning approaches, which combine multiple models, can enhance the accuracy and reliability of predictions.



FIGURE 17. Orange keywords community.



FIGURE 18. Red keywords community.

The orange community's focus on skewed data and class imbalance is particularly noteworthy, as it highlights the challenges and opportunities that arise when dealing with imbalanced datasets. In HR management, skewed data can occur when there are more instances of one class (e.g., employees who have left the company) compared to another class (e.g., employees who are still with the company). By developing techniques to address skewed data and class imbalance, the orange community can create more effective machine learning models that accurately predict HR outcomes. The potential benefits of this research are vast, including improved staff retention, enhanced talent acquisition, and more informed HR decision-making [100].

4) AI AND HR IN DECISION-MAKING

Figure 18 depicts a network of interconnected concepts related to AI, machine learning, and big data. Key elements like AI, Machine Learning, and Big Data are prominently

featured. Each of these elements is crucial for harnessing the power of data and extracting insights that can inform business decisions. The presence of subjects like Human Resource Management (HRM), intellectual capital, and interpretability further highlights the group's focus on applying AI to support HR tasks.

The group's emphasis on HRM and intellectual capital suggests a keen interest in leveraging AI to streamline HR processes and unlock the full potential of intellectual capital within organizations. By harnessing the power of AI, HR professionals can gain strategic insights that inform talent management, employee engagement, and overall business performance. The incorporation of interpretability also indicates a desire to ensure that AI-driven insights are transparent, explainable, and actionable, enabling HR professionals to make informed decisions that drive business outcomes. Overall, Figure 18 provides a compelling glimpse into the group's ambition to harness the power of AI to revolutionize HR practices and drive business success. [17].

V. DISCUSSION

This study examined the co-authorship network within AI applications in HRM research using social network analysis. The findings provide insights into the key players, collaboration patterns, and emerging themes in this rapidly evolving field. Here we discuss the results in relation to the four research questions posed:

Q1)Who are the most influential authors in the HRM and AI co-authoring network?

The analysis identified several highly influential authors based on centrality measures, with Pedrycz Witold, Chen C.L. Philip, and Wang Wei emerging as the top three most central authors. These researchers demonstrate strong collaborative ties and play pivotal roles in connecting different research communities within the AI-HRM domain. Their prominence suggests they are driving much of the innovative work bridging AI and HRM.

Interestingly, while these top authors come from diverse geographical backgrounds (Canada, China, and the USA respectively), there is a notable concentration of influential researchers from China and other Asian countries. This aligns with the rapid growth of AI research and implementation in these regions. The diversity of top authors' expertise - ranging from computational intelligence to data mining and strategic management - highlights the interdisciplinary nature of AI applications in HRM.

Q2) Is the collaboration between countries in coauthorship driven by shared challenges, and what underpins scientific collaboration among universities and institutions in HRM and AI?

The analysis revealed distinct collaboration communities among countries, suggesting that shared regional challenges and research priorities do influence co-authorship patterns. For instance, the "Global HR Applications" community centered around European countries indicates a common focus on integrating AI into HR practices within the EU regulatory framework. Similarly, the "HRM in the Middle East and Asia" community reflects shared interests in leveraging AI for HR in emerging economies.

Institutional collaboration appears to be driven by several factors:

- Geographic proximity, as seen in the clustering of Asian research centers
- Shared research interests, exemplified by the community focused on AI in higher education
- Access to resources and expertise, with leading AI research institutions forming collaborative networks

These collaboration patterns suggest that while global cooperation exists, there are still regional and institutional clusters that may benefit from increased cross-pollination of ideas.

Q3) What underpins scientific collaboration among universities and institutions in HRM and AI?

The analysis identified five main communities in the institutional co-authorship network, each with distinct characteristics:

- AI Research Centers Dominant in Asia: Focused on adapting Western HRM practices and integrating AI
- AI in Leading US Universities: Addressing labor shortages and virtual team management
- Leaders in AI Research in Europe: Emphasizing more informal, culturally-aligned HRM practices
- AI in Higher Education on a Global Scale: Exploring cultural variations in AI-driven HRM
- Newly Established AI Research Centers: Representing a mix of developing and progressing economies

These communities reflect different priorities and approaches to AI in HRM, influenced by factors such as cultural context, economic development stage, and specific organizational challenges. The diversity of these communities underscores the need for contextualized approaches to AI implementation in HRM across different global regions.

Q4) Does the analysis of keyword networks in HRM and AI research reveal distinct and coherent patterns?

The keyword network analysis revealed four distinct communities, each representing a coherent research theme:

- 1. AI for System Identification and Control: Focusing on neural networks and adaptive control in HR processes
- 2. HR Analytics and Performance Management: Emphasizing the use of AI for enhancing HR processes, particularly in SMEs
- 3. Machine Learning for Classification and Prediction: Exploring applications like identifying high-potential staff or predicting turnover
- 4. AI and HR in Decision-Making: Concentrating on the integration of AI, machine learning, and big data for HR decision support

These themes provide a clear roadmap of the current focus areas in AI-HRM research. They suggest that the field is progressing from basic implementation of AI tools to more sophisticated applications that can potentially transform HR decision-making and strategic planning.

The emergence of these distinct yet interrelated themes indicates that AI in HRM is a multifaceted field, with researchers exploring various aspects from technical implementation to strategic impact. This diversity of research focus areas suggests a rich and evolving landscape of AI applications in HRM, with ample opportunities for further innovation and cross-pollination of ideas across these thematic communities.

A. MANAGERIAL INSIGHTS AND IMPLICATION

The findings from this social network analysis of AI applications in HRM research offer valuable insights for managers and practitioners seeking to leverage AI technologies in their HR practices. These insights can guide implementation strategies and help organizations navigate the complex landscape of AI-driven HRM.

1) STRATEGIC COLLABORATION AND KNOWLEDGE SHARING

The identification of key influential authors and institutions provides a roadmap for managers looking to stay at the forefront of AI-HRM innovations. Managers should:

- Actively seek collaborations or partnerships with leading institutions and researchers in the field.
- Encourage their HR teams to engage with academic research, perhaps through industry-academia partnerships or sponsored research programs.
- Implement knowledge-sharing mechanisms within their organizations to disseminate insights from cutting-edge research.

2) REGIONAL ADAPTATION OF AI-HRM PRACTICES

The analysis revealed distinct regional patterns in AI-HRM research, suggesting that one-size-fits-all approaches may not be optimal. Managers should:

- Tailor AI-HRM implementations to local contexts, considering cultural nuances, regulatory environments, and regional challenges.
- Look to region-specific research clusters for insights that may be most relevant to their operational context.
- Balance global best practices with localized approaches, especially in multinational corporations.

3) FOCUS ON EMERGING THEMES

The keyword analysis highlighted four main themes in AI-HRM research. Managers can use these to prioritize their AI initiatives:

a: AI FOR SYSTEM IDENTIFICATION AND CONTROL

This theme focuses on leveraging AI technologies for systematic workforce management and organizational control mechanisms:

- Workforce Planning Systems: Advanced AI algorithms analyze historical data, market trends, and organizational requirements to optimize resource allocation and workforce distribution. These systems can predict staffing needs across departments and projects with greater accuracy.
- Adaptive Performance Management: Implementation of real-time monitoring and feedback systems that automatically adjust performance metrics based on changing business conditions. This includes dynamic goal-setting and performance evaluation frameworks that evolve with organizational needs.
- Competency Gap Analysis: AI-powered systems that continuously assess employee skills against evolving job requirements, identifying training needs and development opportunities proactively. These systems can predict future skill requirements and suggest targeted learning interventions.

b: HR ANALYTICS AND PERFORMANCE MANAGEMENT

This theme encompasses the integration of advanced analytics into core HR functions:

- Data Infrastructure Development: Implementation of robust data collection and storage systems capable of handling diverse HR data types, from performance metrics to employee feedback. This includes establishing data governance frameworks and ensuring data quality.
- Real-time Performance Analytics: Development of sophisticated dashboards that provide instant insights into employee performance, team dynamics, and organizational health metrics. These systems enable proactive intervention and continuous improvement.
- Personalized Development Tracking: AI-driven platforms that create and monitor individualized employee development plans, tracking progress and automatically adjusting learning paths based on performance data and career aspirations.

c: MACHINE LEARNING FOR CLASSIFICATION AND PREDICTION

This theme focuses on applying ML algorithms to enhance HR decision-making processes:

- Advanced Talent Acquisition: Implementation of sophisticated ML models that analyze candidate profiles, job requirements, and organizational culture to predict job fit and success probability. These systems consider both technical skills and soft competencies.
- Retention Risk Analytics: Development of predictive models that identify flight risks by analyzing patterns in employee behavior, engagement levels, and

external market conditions. These systems enable proactive retention strategies.

• High-potential Identification: ML algorithms that analyze multiple data points (performance history, leadership qualities, learning agility) to identify and track high-potential employees for succession planning.

d: AI AND HR DECISION-MAKING

This theme explores the integration of AI into strategic HR decision processes:

- Strategic Planning Support: Implementation of AI systems that analyze workforce trends, market conditions, and organizational data to support strategic HR planning and policy development. These systems help identify long-term workforce needs and challenges.
- Advanced Sentiment Analysis: Development of sophisticated NLP systems that analyze employee feedback, communication patterns, and social interactions to gauge organizational climate and employee satisfaction levels.
- Objective Decision Frameworks: Implementation of AI-powered decision support systems that reduce bias in HR processes by standardizing evaluation criteria and providing data-driven recommendations for promotions, compensation, and career advancement.

4) INTERDISCIPLINARY APPROACH

The diversity of backgrounds among top researchers suggests that effective AI-HRM implementation requires an interdisciplinary approach. Managers should:

- Form cross-functional teams that include HR professionals, data scientists, and IT specialists.
- Invest in upskilling HR staff in data literacy and basic AI concepts.
- Foster a culture of continuous learning and adaptation to keep pace with AI advancements.

5) BRIDGING ACADEMIA AND INDUSTRY

The strong presence of academic institutions in the network suggests a wealth of theoretical knowledge that may not yet be fully translated into practice. Managers should:

- Establish partnerships with universities for applied research in AI-HRM.
- Offer internships or collaborative projects to bridge the gap between academic research and practical implementation.
- Contribute to academic research by sharing real-world implementation challenges and successes.

the successful implementation of AI in HRM requires a strategic, multifaceted approach. By leveraging insights from the research landscape, fostering collaborations, and maintaining a balance between innovation and ethical considerations, managers can position their organizations to fully harness the potential of AI in HRM. The key lies in viewing AI not as a replacement for human judgment, but as a powerful tool to augment and enhance HR capabilities, ultimately leading to more data-driven, efficient, and employee-centric HR practices.

B. LIMITATIONS AND FUTURE RESEARCH

Despite the valuable insights provided by this study, there are several limitations to consider. First, the analysis relies primarily on co-authorship networks and citation data from the Web of Science database, which may not fully capture the breadth of research activities, particularly from grey literature or non-English publications. Additionally, the study's focus on quantitative metrics such as centrality measures and TOPSIS may overlook qualitative aspects of collaboration dynamics, such as the depth of research partnerships or the nature of interdisciplinary work. Finally, the dataset, while extensive, may have limitations in terms of representing emerging trends in AI-HRM research, as newer works may not yet be fully captured in citation data.

Future research could address the limitations of this study by incorporating alternative sources of data, such as patents, conference proceedings, or industry reports, to provide a more comprehensive view of AI applications in HRM. Furthermore, qualitative studies exploring the nature of collaboration, including interviews with key authors and institutions, could provide deeper insights into the mechanisms of research partnerships. In addition, expanding the focus to include the practical applications of AI in HRM within organizations and its impact on workforce outcomes would enrich the understanding of AI's role in HRM. Longitudinal studies tracking the evolution of AI-HRM integration over time would also be valuable to capture the dynamic changes in the field and their implications for HR practices.

VI. CONCLUSION

This study provides a comprehensive analysis of the research landscape in AI-HRM using social network analysis. By examining co-authorship networks, institutional collaborations, and keyword patterns, this research offers valuable insights into the structure and dynamics of this rapidly evolving field.

The analysis reveals a complex and interconnected research ecosystem, with several key findings:

- **Influential Authors:** The study identifies a core group of highly influential researchers, led by Pedrycz Witold, Chen C.L. Philip, and Wang Wei, who play pivotal roles in shaping the field. These authors, from diverse geographical and academic backgrounds, demonstrate the interdisciplinary nature of AI-HRM research and highlight the global reach of this emerging field.
- Global Collaboration Patterns: The research uncovers distinct collaboration communities among countries, reflecting shared regional challenges and research priorities. The identification of three main communities -Global HR Applications, HRM in the Middle East and Asia, and Global Integration of AI in HRM - illustrates

the diverse approaches to AI-HRM across different global regions and emphasizes the importance of contextual factors in shaping research agendas.

- **Institutional Networks:**The analysis of institutional co-authorship reveals five distinct communities, each with unique characteristics and focus areas. This finding underscores the varied approaches to AI-HRM research across different academic and cultural contexts, from the adaptation of Western HRM practices in Asia to the exploration of AI in higher education globally.
- Emerging Research Themes: The keyword network analysis identifies four primary research themes: AI for System Identification and Control, HR Analytics and Performance Management, Machine Learning for Classification and Prediction, and AI and HR in Decision-Making. These themes provide a roadmap for future research and highlight the multifaceted nature of AI applications in HRM.

The study's findings have significant implications for both academia and industry:

- For researchers, this study provides a comprehensive overview of the current state of AI-HRM research, identifying key players, collaboration opportunities, and emerging themes. This can guide future research directions and foster more targeted collaborations across institutions and regions.
- **For practitioners**, the identified research themes offer insights into the most promising areas for AI application in HRM. This can help organizations prioritize their AI investments and implementation efforts, aligning their initiatives with cutting-edge research.
- For policymakers, This study offers valuable insights for policymakers by elucidating the impact of AI-driven human resource management (HRM) practices on organizational efficiency and workforce management. By mapping the emerging trends and collaboration patterns in AI-HRM research, policymakers can gain a clearer understanding of the potential benefits and challenges associated with integrating AI in HRM practices. These insights can guide the development of regulatory frameworks and public policies that support the responsible and effective adoption of AI technologies, ensuring that AI integration in HRM aligns with national workforce development goals, improves productivity, and mitigates potential risks such as job displacement or skill gaps.

In conclusion, this study offers a thorough and insightful mapping of the AI-HRM research landscape, shedding light on the prevailing trends, influential contributors, and emerging directions in the field. As artificial intelligence progressively reshapes human resource management practices, this research serves as a crucial resource for scholars, practitioners, and policymakers. For policymakers, the findings guide the formulation of strategies that foster international collaboration by pinpointing key research hubs and partnerships. Furthermore, the study identifies critical areas where AI implementation in HRM necessitates careful consideration of ethical and regulatory issues. It underscores the importance of ensuring responsible deployment of AI technologies, promoting innovation while addressing challenges related to bias, fairness, and privacy. Ultimately, this study lays the groundwork for navigating the dynamic intersection of AI and HRM, encouraging global collaboration and responsible AI integration in HR practices. Future research should expand on these findings, delving into the qualitative aspects of AI-HRM implementation, exploring ethical dilemmas, and evaluating the long-term implications of AI on workforce dynamics and organizational performance. Additionally, longitudinal studies are essential to track the evolution of this field, providing a deeper understanding of how AI-HRM research and practice continue to shape the future of work.

REFERENCES

- R. Deepa, S. Sekar, A. Malik, J. Kumar, and R. Attri, "Impact of AI-focussed technologies on social and technical competencies for HR managers—A systematic review and research agenda," *Technol. Forecasting Social Change*, vol. 202, May 2024, Art. no. 123301.
- [2] E. P. Kostenko, "Transformation of the role and functions of HR in the conditions of new economic challenges," *J. Econ. Regulation*, vol. 12, no. 4, pp. 94–109, Dec. 2021.
- [3] N. Geethanjali, K. Ashifa, A. Raina, J. Patil, R. Byloppilly, and S. S. Rajest, "Application of strategic human resource management models for organizational performance," in *Data-Driven Decision Making for Long-Term Bus. Success.* Hershey, PA, USA: IGI Global, 2024, pp. 1–19.
- [4] O. Hridin, "General trends and characteristic aspects of digital transformation in the field of HR management," *Eastern Europe, Economy, Business Manage.*, no. 3, p. 40, Jan. 2023.
- [5] C. Sharma and N. Chanana, "AI in HRM: Unveiling areas and future trajectories for organizational transformation," in *Proc. ASU Int. Conf. Emerg. Technol. Sustainability Intell. Syst. (ICETSIS)*, Jan. 2024, pp. 1736–1740.
- [6] E. E. Bennett and R. R. McWhorter, "Virtual HRD's role in crisis and the post COVID-19 professional lifeworld: Accelerating skills for digital transformation," *Adv. Developing Hum. Resour.*, vol. 23, no. 1, pp. 5–25, Feb. 2021.
- [7] H. Xiang, J. Lu, M. E. Kosov, M. V. Volkova, V. V. Ponkratov, A. I. Masterov, I. D. Elyakova, S. Y. Popkov, D. Y. Taburov, N. V. Lazareva, I. Muda, M. V. Vasiljeva, and A. O. Zekiy, "Sustainable development of employee lifecycle management in the age of global challenges: Evidence from China, Russia, and Indonesia," *Sustainability*, vol. 15, no. 6, p. 4987, Mar. 2023.
- [8] D. Dutta and A. Kannan Poyil, "The machine/human agentic impact on practices in learning and development: A study across MSME, NGO and MNC organizations," *Personnel Rev.*, vol. 53, no. 3, pp. 791–815, May 2024.
- [9] G. Kaur, R. Gujrati, and H. Uygun, "How does AI fit into the management of human resources?" *Rev. Artif. Intell. Educ.*, vol. 4, p. e04, Aug. 2023.
- [10] P. V. Yadav, U. S. Kollimath, S. A. Giramkar, D. T. Pisal, S. S. Badave, and S. M. Swamy, "Impact of artificial intelligence (AI) on various aspects of human resources management (HRM) in India," in *Proc. 3rd Int. Conf. Emerg. Smart Technol. Appl. (eSmarTA)*, Oct. 2023, pp. 1–6.
- [11] M. N. I. Afzal, A. H. N. Shohan, S. Siddiqui, and N. Tasnim, "Application of AI on human resource management: A review," *J. Hum. Resource Manage.-HR Adv. Develop.*, vol. 2023, no. 1, pp. 1–11, Aug. 2023.
- [12] S. Majumder and A. Mondal, "Are chatbots really useful for human resource management?" *Int. J. Speech Technol.*, vol. 24, no. 4, pp. 969–977, Dec. 2021.
- [13] N. Kaushal, R. P. S. Kaurav, B. Sivathanu, and N. Kaushik, "Artificial intelligence and HRM: Identifying future research agenda using systematic literature review and bibliometric analysis," *Manage. Rev. Quart.*, vol. 73, no. 2, pp. 455–493, Jun. 2023.

- [14] P. R. Palos-Sánchez, P. Baena-Luna, A. Badicu, and J. C. Infante-Moro, "Artificial intelligence and human resources management: A bibliometric analysis," *Appl. Artif. Intell.*, vol. 36, no. 1, Dec. 2022, Art. no. 2145631.
- [15] B. O. Saracoglu, "Initialization of profile and social network analyses robot and platform with a concise systematic review," *Mach. Learn. Appl.*, vol. 7, Mar. 2022, Art. no. 100249.
- [16] Q. Wang, "The impact of AI on organizational employees: A literature review," J. Educ., Humanities Social Sci., vol. 19, pp. 45–53, Aug. 2023.
- [17] P. Budhwar, A. Malik, M. T. De Silva, and P. Thevisuthan, "Artificial intelligence—Challenges and opportunities for international HRM: A review and research agenda," *Int. J. Hum. Resour. Manage.*, vol. 33, no. 6, pp. 1065–1097, 2022.
- [18] S. Saleh and A. Gajendran, "Catalyzing organizational development and enhancing employee satisfaction: The transformative influence of artificial intelligence software," *Int. J. Sci. Res. Eng. Manage.*, vol. 7, no. 10, pp. 1–11, Oct. 2023.
- [19] R. Tsiskaridze, K. Reinhold, and M. Jarvis, "Innovating HRM recruitment: A comprehensive review of AI deployment," *Marketing Manage. Innov.*, vol. 14, no. 4, pp. 239–254, 2023.
- [20] N. Böhmer and H. Schinnenburg, "Critical exploration of AI-driven HRM to build up organizational capabilities," *Employee Relations, Int. J.*, vol. 45, no. 5, pp. 1057–1082, Jul. 2023.
- [21] V. Prikshat, M. Islam, P. Patel, A. Malik, P. Budhwar, and S. Gupta, "AIaugmented HRM: Literature review and a proposed multilevel framework for future research," *Technol. Forecasting Social Change*, vol. 193, Aug. 2023, Art. no. 122645.
- [22] A. H. Fomude, C. Yang, G. K. Agordzo, A. V. Serwah, and L. Abangbila, "AI model to improve HR decision-making with machine learning predictions algorithm," in *Proc. 25th Int. Conf. Adv. Commun. Tech*nol. (ICACT), Feb. 2023, pp. 206–212.
- [23] R. Khan, A. Hussain, and S. Ahmad, "Revolutionizing human resource management: The transformative impact of artificial intelligence (AI) applications," *Int. J. Social Sci. Entrepreneurship*, vol. 3, no. 4, pp. 306–326, Oct. 2023.
- [24] W. Gryncewicz, R. Zygała, and A. Pilch, "AI in HRM: Case study analysis. Preliminary research," *Proc. Comput. Sci.*, vol. 225, pp. 2351–2360, Jan. 2023, doi: 10.1016/j.procs.2023.10.226.
- [25] M. Goswami, S. Jain, T. Alam, A. F. Deifalla, A. E. Ragab, and R. Khargotra, "Exploring the antecedents of AI adoption for effective HRM practices in the Indian pharmaceutical sector," *Frontiers Pharma*col., vol. 14, Nov. 2023.
- [26] S. Bhakuni, "Application of artificial intelligence on human resource management in information technolgy industry in India," *Sci. Temper*, vol. 14, no. 4, pp. 1232–1243, Dec. 2023.
- [27] A. Jatobá and P. V. R. D. Carvalho, "Resilience in public health: Precepts, concepts, challenges, and perspectives," *Saúde em Debate*, vol. 46, no. spe8, pp. 130–139, Dec. 2022.
- [28] E. Kambur and T. Yildirim, "From traditional to smart human resources management," *Int. J. Manpower*, vol. 44, no. 3, pp. 422–452, May 2023.
- [29] P. Malik, K. Patel, C. Pinto, R. Jaiswal, R. Tirupathi, S. Pillai, and U. Patel, "Post-acute COVID-19 syndrome (PCS) and health-related quality of life (HRQoL)—A systematic review and meta-analysis," *J. Med. Virol.*, vol. 94, no. 1, pp. 253–262, Jan. 2022.
- [30] A. Malik, P. Budhwar, and B. A. Kazmi, "Artificial intelligence (AI)-assisted HRM: Towards an extended strategic framework," *Hum. Resource Manage. Rev.*, vol. 33, no. 1, Mar. 2023, Art. no. 100940.
- [31] S. Strohmeier and F. Piazza, *Human Resource Intelligence Und Analytics*. Cham, Switzerland: Springer, 2015.
- [32] S. Nimmagadda, R. K. Surapaneni, and R. M. Potluri, "Artificial intelligence in HR: Employee engagement using chatbots," in *Artificial Intelligence Enabled Management: An Emerging Economy Perspective*, 2024, p. 147.
- [33] B. A. M. Johnson, J. D. Coggburn, and J. J. Llorens, "Artificial intelligence and public human resource management: Questions for research and practice," *Public Personnel Manage.*, vol. 51, no. 4, pp. 538–562, Dec. 2022.
- [34] Q. Hu, "Unilever's practice on AI-based recruitment," *Highlights Bus.*, *Econ. Manage.*, vol. 16, pp. 256–263, Aug. 2023.
- [35] R. Nyathani, "AI-powered recruitment the future of HR digital transformation," J. Artif. Intell. Cloud Comput., vol. 2022, pp. 1–5, Dec. 2022.

- [36] S. Basu, B. Majumdar, K. Mukherjee, S. Munjal, and C. Palaksha, "Artificial intelligence–HRM interactions and outcomes: A systematic review and causal configurational explanation," *Hum. Resource Manage. Rev.*, vol. 33, no. 1, Mar. 2023, Art. no. 100893.
- [37] A. Thakur, P. Hinge, and V. R. Adhegaonkar, "Use of artificial intelligence (AI) in recruitment and selection," in *Proc. Int. Conf. Appl. Mach. Intell. Data Anal.*, Jan. 2023.
- [38] Dr. S. V. Dr. Bijja Vishwanath, "The future of work: Implications of artificial intelligence on hr practices," *Tuijin Jishu/J. Propuls. Technol.*, vol. 44, no. 3, pp. 1711–1724, Oct. 2023.
- [39] S. H. H. Mazloumi, A. Moini, and M. A. M. A. Kermani, "Designing synchronizer module in CMMS software based on lean smart maintenance and process mining," *J. Quality Maintenance Eng.*, vol. 29, no. 2, pp. 509–529, Apr. 2023, doi: 10.1108/jqme-10-2021-0077.
- [40] J. J. G. Hernández, G. M. Díaz, and J. L. G. Salvador, "Artificial intelligence applied to human resources management: A bibliometric analysis," in *Proc. Int. Conf. Inf. Technol. Syst.* Cham, Switzerland: Springer, Jan. 2024, pp. 269–277.
- [41] V. Singh and A. Khatun, "Application of artificial intelligence in human resource management: A conceptual framework," in *The Role of HR in the Transforming Workplace*. Productivity Press, 2024, pp. 32–49.
- [42] K. Ekuma, "Artificial intelligence and automation in human resource development: A systematic review," *Hum. Resource Develop. Rev.*, vol. 23, no. 2, pp. 199–229, Jun. 2024.
- [43] T. Ć ormarković, L. Dražeta, and A. Njeguš, "The levels of artificial intelligence application in human resource systems," *Eur. J. Appl. Econ.*, vol. 19, no. 2, pp. 28–42, 2022.
- [44] L. Yan and W. Zhiping, "Mapping the literature on academic publishing: A bibliometric analysis on WOS," *Sage Open*, vol. 13, no. 1, Jan. 2023, Art. no. 21582440231158562.
- [45] Z. Chinchilla-Rodríguez, R. Costas, N. Robinson-García, and V. Larivière, "Examining the quality of the corresponding authorship field in web of science and scopus," *Quant. Sci. Stud.*, vol. 5, no. 1, pp. 76–97, Mar. 2024.
- [46] I. Mitrović, M. Mišić, and J. Protić, "Exploring high scientific productivity in international co-authorship of a small developing country based on collaboration patterns," J. Big Data, vol. 10, no. 1, p. 64, May 2023.
- [47] J. P. Andersen, "Field-level differences in paper and author characteristics across all fields of science in web of science, 2000–2020," *Quantum Sci. Stud.*, vol. 4, no. 2, pp. 394–422, Jan. 2023.
- [48] A. Sarkar, H. Wang, A. Rahman, W. H. Memon, and L. Qian, "A bibliometric analysis of sustainable agriculture: Based on the web of science (WOS) platform," *Environ. Sci. Pollut. Res.*, vol. 29, no. 26, pp. 38928–38949, Jun. 2022.
- [49] L. Xu, H. Liu, X. Yuan, E. Chen, and Y. Tang, "GraKerformer: A transformer with graph kernel for unsupervised graph representation learning," *IEEE Trans. Cybern.*, vol. 54, no. 12, pp. 7320–7332, Dec. 2024, doi: 10.1109/tcyb.2024.3465213.
- [50] H. Li and W. Liu, "Same same but different: Self-citations identified through scopus and web of science core collection," *Scientometrics*, vol. 124, no. 3, pp. 2723–2732, Sep. 2020.
- [51] M. Maghsoudi, S. Shokouhyar, A. Ataei, S. Ahmadi, and S. Shokoohyar, "Co-authorship network analysis of AI applications in sustainable supply chains: Key players and themes," *J. Cleaner Prod.*, vol. 422, Oct. 2023, Art. no. 138472, doi: 10.1016/j.jclepro.2023.138472.
- [52] C. Zhang, H. Huang, C. Shi, J. Xu, and Y.-H. Chiu, "Spatial pattern and driving factors for interprovincial water use in China: Based on SNA and LMDI," *Energy Environ.*, vol. 35, no. 4, pp. 2198–2227, Jun. 2024.
- [53] Q. Liu, R. Chen, and Q. Yu, "Improving sustainable performance of China's new energy industry through collaborative innovation network resilience," *Renew. Sustain. Energy Rev.*, vol. 201, Sep. 2024, Art. no. 114625.
- [54] Y. Qin and R. Chen, "Social network analysis of COVID-19 research and the changing international collaboration structure," J. Shanghai Jiaotong Univ., Sci., vol. 29, no. 1, pp. 150–160, Feb. 2024.
- [55] M. Kermani, A. Karimimajd, N. Mohammadi, and B. Aliahmadi, "A note on predicting how people interact in attributed social networks," *Int. J. Curr. Life Sci.*, vol. 4, pp. 2510–2514, 2014.
- [56] L. C. Smit, J. Dikken, M. J. Schuurmans, N. J. de Wit, and N. Bleijenberg, "Value of social network analysis for developing and evaluating complex healthcare interventions: A scoping review," *BMJ Open*, vol. 10, no. 11, Nov. 2020, Art. no. e039681.

- [57] M. Maghsoudi, M. Jalilvand Khosravi, and S. Salavatian, "Analyzing the ecosystem of Iranian websites for watching and downloading videos based on user behavior," *Iranian J. Inf. Process. Manage.*, vol. 38, no. 3, pp. 1067–1094, 2023.
- [58] M. Maghsoudi, "Uncovering the skillsets required in computer science jobs using social network analysis," *Educ. Inf. Technol.*, vol. 29, no. 10, pp. 12759–12780, Jul. 2024, doi: 10.1007/s10639-023-12304-4.
- [59] A. I. Bento, C. Cruz, G. Fernandes, and L. M. D. F. Ferreira, "Social network analysis: Applications and new metrics for supply chain management—A literature review," *Logistics*, vol. 8, no. 1, p. 15, Feb. 2024.
- [60] T. Prochnow and M. S. Patterson, "Assessing social network influences on adult physical activity using social network analysis: A systematic review," *Amer. J. Health Promotion*, vol. 36, no. 3, pp. 537–558, Mar. 2022.
- [61] F. García-Lillo, P. Seva-Larrosa, and E. Sánchez-García, "What is going on in entrepreneurship research? A bibliometric and SNA analysis," *J. Bus. Res.*, vol. 158, Mar. 2023, Art. no. 113624.
- [62] E. Lee and G. Kim, "A comparative study on critical legal issues in Korea and EU countries on climate change litigations: Based on social network analysis (SNA)," in *Judicial Responses to Climate Change in the Global South: A Jurisdictional and Thematic Review*, 2023, pp. 141–163.
- [63] A. Noroozian, B. Amiri, and S. Ramezani, "In the arena of the content war: A social network analysis approach for content differentiation in VOD platforms," *IEEE Access*, vol. 12, pp. 81137–81155, 2024.
- [64] M. Saqr, S. López-Pernas, M. Á. Conde-González, and Á. Hernández-García, "Social network analysis: A primer, a guide and a tutorial in R," in *Learning Analytics Methods and Tutorials: A Practical Guide Using R*, 2024, pp. 491–518.
- [65] L. Xu, Z. Wang, S. Zhang, X. Yuan, M. Wang, and E. Chen, "Modeling student performance using feature crosses information for knowledge tracing," *IEEE Trans. Learn. Technol.*, vol. 17, pp. 1364–1377, 2024, doi: 10.1109/TLT.2024.3381045.
- [66] N. Bakkalbasi and T. Krichel, "Patterns of research collaboration in a digital library for economics," *Proc. Amer. Soc. Inf. Sci. Technol.*, vol. 43, no. 1, pp. 1–15, Jan. 2006.
- [67] B. Amiri and R. Karimianghadim, "A novel text clustering model based on topic modelling and social network analysis," *Chaos, Solitons Fractals*, vol. 181, Apr. 2024, Art. no. 114633.
- [68] C. Zapelini, D. S. de Souza, R. O. de Santana, J. Olaya-Restrepo, and A. Schiavetti, "Environmental, social, and management aspects in a hotspot: Interaction networks between marine protected areas," *Ocean Coastal Manage.*, vol. 251, May 2024, Art. no. 107068.
- [69] A. H. Ganai, R. Hashmy, and H. A. Khanday, "Finding information Diffusion's seed nodes in online social networks using a special degree centrality," *Social Netw. Comput. Sci.*, vol. 5, no. 4, p. 333, 2024.
- [70] K. Shabankareh, A. Hamidi, M. R. Soleymani, H. S. Sajadi, and M. Alavi, "Faculty members as academic knowledge brokers in Iran's health sector: A social network analysis study," *Health Res. Policy Syst.*, vol. 22, no. 1, p. 53, Apr. 2024.
- [71] A. Noroozian, B. Amiri, and M. Agha Mohammad Ali Kermani, "Investigation of cinematic genre diversity based on social network analysis: The lost ring of the Iranian cinema industry," *Kybernetes*, vol. 52, no. 12, pp. 6321–6347, Nov. 2023.
- [72] H. Baziyad, V. Kayvanfar, and M. Toloo, "A data envelopment analysis model for opinion leaders' identification in social networks," *Comput. Ind. Eng.*, vol. 190, Apr. 2024, Art. no. 110010.
- [73] Q. Zhang, R. Deng, K. Ding, and M. Li, "Structural analysis and the sum of nodes' betweenness centrality in complex networks," *Chaos, Solitons Fractals*, vol. 185, Aug. 2024, Art. no. 115158.
- [74] M. A. Fallah, M. Agha Mohammad Ali Kermani, A. Moini, and J. Mashayekh, "Identification and evaluation of collaborative relationships of internal project stakeholders: A social network analysis approach," *Kybernetes*, vol. 53, no. 10, pp. 3621–3658, Oct. 2024.
- [75] P. Barroso and C. Gortázar, "New insights into biodiversity-disease relationships: The importance of the host community network characterization," *Eur. J. Wildlife Res.*, vol. 70, no. 3, p. 48, Jun. 2024.
- [76] S. H. H. Mazlumi and M. A. M. Kermani, "Investigating the structure of the Internet of Things patent network using social network analysis," *IEEE Internet Things J.*, vol. 9, no. 15, pp. 13458–13469, Aug. 2022.
- [77] R. M. D'Souza, M. di Bernardo, and Y.-Y. Liu, "Controlling complex networks with complex nodes," *Nature Rev. Phys.*, vol. 5, no. 4, pp. 250–262, Mar. 2023.

- [78] M. Reza HabibAgahi, M. Agha Mohammad Ali Kermani, and M. Maghsoudi, "On the co-authorship network analysis in the process mining research community: A social network analysis perspective," *Expert Syst. Appl.*, vol. 206, Nov. 2022, Art. no. 117853.
- [79] M. Agha Mohammad Ali Kermani, A. Aliahmadi, and R. Hanneman, "Optimizing the choice of influential nodes for diffusion on a social network," *Int. J. Commun. Syst.*, vol. 29, no. 7, pp. 1235–1250, May 2016.
- [80] J. El-Ouahi, "Research funding in the middle east and North Africa: Analyses of acknowledgments in scientific publications indexed in the web of science (2008–2021)," *Scientometrics*, vol. 129, no. 6, pp. 2933–2968, May 2024.
- [81] A. L. Rossoni, E. P. G. de Vasconcellos, and R. L. de Castilho Rossoni, "Barriers and facilitators of university-industry collaboration for research, development and innovation: A systematic review," *Manage. Rev. Quart.*, vol. 74, no. 3, pp. 1841–1877, Sep. 2024.
- [82] J. Knox, "Artificial intelligence and education in China," *Learn., Media Technol.*, vol. 45, no. 3, pp. 298–311, Apr. 2020.
- [83] F. L. Cooke, R. Schuler, and A. Varma, "Human resource management research and practice in Asia: Past, present and future," *Hum. Resource Manage. Rev.*, vol. 30, no. 4, Dec. 2020, Art. no. 100778.
- [84] S. Tehseen, S. Khalid, R. A. Rather, Z. H. Qureshi, and H. A. Halbusi, "HRM practices for knowledge management and retail firms performances: A comparative study among Malay and Chinese firms," *Int. J. Entrepreneurship*, vol. 24, no. 1, Apr. 2020.
- [85] M. Warner and A. Nankervis, "HRM practices in Chinese MNCs: Rhetoric and reality," J. Gen. Manage., vol. 37, no. 3, pp. 61–80, Mar. 2012.
- [86] A. Bhutoria, "Personalized education and artificial intelligence in the United States, China, and India: A systematic review using a humanin-the-loop model," *Comput. Educ., Artif. Intell.*, vol. 3, Jan. 2022, Art. no. 100068, doi: 10.1016/j.caeai.2022.100068.
- [87] B. George and O. Wooden, "Managing the strategic transformation of higher education through artificial intelligence," *Administ. Sci.*, vol. 13, no. 9, p. 196, 2023.
- [88] V. Pereira, E. Hadjielias, M. Christofi, and D. Vrontis, "A systematic literature review on the impact of artificial intelligence on workplace outcomes: A multi-process perspective," *Hum. Resource Manage. Rev.*, vol. 33, no. 1, Mar. 2023, Art. no. 100857.
- [89] O. Popo-Olaniyan, O. A. Elufioye, F. C. Okonkwo, C. A. Udeh, T. F. Eleogu, and F. O. Olatoye, "AI-driven talent analytics for strategic HR decision-making in the United States of America: A review," *Int. J. Manage. Entrepreneurship Res.*, vol. 4, no. 12, pp. 607–622, Dec. 2023.
- [90] A. Arslan, C. Cooper, Z. Khan, I. Golgeci, and I. Ali, "Artificial intelligence and human workers interaction at team level: A conceptual assessment of the challenges and potential HRM strategies," *Int. J. Manpower*, vol. 43, no. 1, pp. 75–88, Apr. 2022.
- [91] M. Mohiuddin, E. Hosseini, S. B. Faradonbeh, and M. Sabokro, "Achieving human resource management sustainability in universities," *Int. J. Environ. Res. Public Health*, vol. 19, no. 2, p. 928, Jan. 2022.
- [92] C. Feijóo, Y. Kwon, J. M. Bauer, E. Bohlin, B. Howell, R. Jain, P. Potgieter, K. Vu, J. Whalley, and J. Xia, "Harnessing artificial intelligence (AI) to increase wellbeing for all: The case for a new technology diplomacy," *Telecommun. Policy*, vol. 44, no. 6, Jul. 2020, Art. no. 101988.
- [93] N. Tomašev et al., "AI for social good: Unlocking the opportunity for positive impact," *Nature Commun.*, vol. 11, no. 1, p. 2468, May 2020.
- [94] L. Hardman, "Artificial intelligence along the new silk road: Competition or collaboration?" in *China and Europe on the New Silk Road*, 2020, pp. 240–254.
- [95] S. Feng and N. Law, "Mapping artificial intelligence in education research: A network-based keyword analysis," *Int. J. Artif. Intell. Educ.*, vol. 31, no. 2, pp. 277–303, Jun. 2021.
- [96] D. F. Andrade, J. P. Romanelli, and E. R. Pereira-Filho, "Past and emerging topics related to electronic waste management: Top countries, trends, and perspectives," *Environ. Sci. Pollut. Res.*, vol. 26, no. 17, pp. 17135–17151, Jun. 2019.
- [97] J.-Y. Ha and H.-J. Park, "Keyword network analysis and topic modeling of news articles related to artificial intelligence and nursing," *J. Korean Acad. Nursing*, vol. 53, no. 1, p. 55, 2023.

- [98] R. Baldegger, M. Caon, and K. Sadiku, "Correlation between entrepreneurship orientation and implementation of AI in human resources management (HRM)," *Technol. Innov. Manage. Rev.*, vol. 10, no. 4, pp. 72–79, Apr. 2020.
- [99] C. Zehir, T. Karaboğa, and D. Bağar, "The transformation of human resource management and its impact on overall business performance: Big data analytics and AI technologies in strategic HRM," in *Digital Busness Strategies in Blockchain Ecosystems: Transformational Design and Future of Global Business*, 2020, pp. 265–279.
- [100] S. Garg, S. Sinha, A. K. Kar, and M. Mani, "A review of machine learning applications in human resource management," *Int. J. Productiv. Perform. Manage.*, vol. 71, no. 5, pp. 1590–1610, May 2022.



MEHRDAD MAGHSOUDI received the master's degree in industrial management from Allameh Tabataba'i University. He is currently pursuing the Ph.D. degree in information technology management with Shahid Beheshti University, Tehran, specializing in business intelligence. His expertise spans a range of advanced techniques, including patent analysis, social network analysis, text mining, and process mining. He has applied these skills in various research projects, showcasing his

ability to integrate complex data analysis methodologies. In addition to his academic work, he serves as a Consultant on data-driven projects, leveraging his knowledge to provide insights and solutions in applied data science and business intelligence.



MOTAHAREH KAMRANI SHAHRI received the degree in information technology management from Iran University of Science and Technology.

She is passionate about data mining, leveraging advanced analytics to uncover patterns and insights, and social network analysis, where she delves into the dynamics of relationships and interactions within organizations. With a strong academic background, she dedicated her master's thesis to exploring the applications of artificial

intelligence in human resource management, where she examined how AI can enhance decision-making processes in HR. Her research reflects a broader interest in the intersection of technology and organizational efficiency. Her work aims to contribute to more effective management strategies in today's data-driven world.



MEHRDAD AGHA MOHAMMAD ALI KER-MANI is currently an Assistant Professor with Iran University of Science and Technology. He holds a specialized Ph.D. degree in industrial engineering, with a focus on the intersection of data analysis and process optimization. His expertise lies in several areas of computer science and data analytics, including process mining, social network analysis, influence maximization, information diffusion, and game theory. His work centers around

developing algorithms and analytical tools to solve complex problems related to these topics. His interest in these subjects began during his Ph.D. studies, where he explored the potential for data analysis techniques to optimize industrial processes. He soon realized that these same techniques could be applied to a wide range of domains, from marketing to public health. Since becoming an Assistant Professor with Iran University of Science and Technology, he has continued to explore these topics through research and teaching. He is passionate about sharing his knowledge with students and colleagues and is committed to advancing their understanding of how data analysis can be used to drive innovation and improve organizational performance. He is well-regarded in his field and has published numerous articles in top-tier academic journals. He is also active in the academic community, regularly presenting his work at conferences and workshops around the world. His contributions have helped to shape the field of data analytics, and he is widely recognized as a thought leader in this area.



RAHIM KHANIZAD received the master's and Ph.D. degrees in information technology. He is a renowned Researcher in information technology, specializing in business intelligence. His work focuses on uncovering insights from large datasets, exploring electronic business models, understanding the digital economy's impact, identifying innovative strategies for technology adoption, and aligning business goals with technology-driven strategies. He has authored numerous publications

and actively shares his expertise in conferences and events. His contributions have greatly influenced academia, industry, and society as a whole. His research interests include data science, electronic business, digital economy, new technologies, and strategic planning.